Author’s response to reviews

Title: Effects of 'plate model' as a part of dietary intervention on rehabilitation of post myocardial infarction patients: study protocol for a randomized controlled trial

Authors:

Ranil Jayawardena (ranil7@gmail.com)
Pasindu Fernando (pasindu9091@gmail.com)
Niroshan Lokunarangoda (niroshan_loku@yahoo.com)
Anidu Pathirana (anidupathirana@hotmail.com)

Version: 2 Date: 16 Jun 2017

Author’s response to reviews:

Editors-in Chief

Trials

Dear Sirs,

RE: Effects of the ‘plate model’ as part of dietary intervention on modification of selected cardio-metabolic risk factors in post myocardial infarction patients: study protocol for a randomized controlled trial.

Thank you very much for inform us that our article is potentially acceptable for publication in Trials. The authors are very grateful to the editor and reviewer for their final comments regarding the improvement of our article. However, there is no significant modifications are requested for this manuscript.

Editor comment:

Has the study started? If not, consider using a better established way of randomization such as sealed envelopes.
Thank you for your comment. Yes, this study is already started and considerable number of the patients have already recruited. We will consider your suggestion on randomization for the future studies.

Reviewer reports:

Reviewer #1: Many thanks for the many changes. I agree with a lot of the changes, however, the research group should be better informed about monitoring, eCRF and the randomization process (according to the GCP guidelines) for the next project.

Thank you for your suggestion. As you suggested we will consider your recommendation in the next project.

Thank you for considering the paper for publication.

Yours sincerely

Dr. Ranil Jayawardena MBBS, MSc, PhD
Faculty of Medicine,
University of Colombo
Sri Lanka